Author's response to reviews

Title: The hospital anxiety and depression rating scale: A cross-sectional study of psychometrics and case finding abilities in general practice

Authors:

Ingrid Olsson (ingrid.olsson@sykehuset-innlandet.no)
Arnstein Mykletun (arnstein.mykletun@psyhp.uib.no)
Alv A Dahl (alvd@ulrik.uio.no)

Version: 6 Date: 13 December 2005

Author's response to reviews:

In the abstract
Under methods: "in three consecutive days" are changed to "on one day"